Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2023GL104894

Key Points:

- The duration of snow-free, bare ice is the primary control on the distribution of dark ice
- Longer durations of bare ice (median = 40 days) are prerequisites for ice to become dark
- Surface slope and aspect influence darkening by modulating the duration of bare ice and glacier ice algae growth

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

S. Feng and M. Tranter, shunan.feng@envs.au.dk; martyn.tranter@envs.au.dk

Citation:

Feng, S., Cook, J. M., Naegeli, K., Anesio, A. M., Benning, L. G., & Tranter, M. (2024). The impact of bare ice duration and geo-topographical factors on the darkening of the Greenland Ice Sheet. *Geophysical Research Letters*, *51*, e2023GL104894. https://doi. org/10.1029/2023GL104894

Received 8 JUN 2023 Accepted 18 DEC 2023

© 2023. The Authors. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

FENG ET AL.

The Impact of Bare Ice Duration and Geo-Topographical Factors on the Darkening of the Greenland Ice Sheet

Shunan Feng¹, Joseph Mitchell Cook¹, Kathrin Naegeli², Alexandre Magno Anesio¹, Liane G. Benning^{3,4}, and Martyn Tranter¹

¹Department of Environmental Science, Aarhus University, Roskilde, Denmark, ²Department of Geography, Remote Sensing Laboratories, University of Zurich, Zurich, Switzerland, ³Section Interface Geochemistry, GFZ German Research Centre for Geosciences, Potsdam, Germany, ⁴Department of Earth Sciences, Free University of Berlin, Berlin, Germany

Abstract Dark (low albedo) surface ice on the Greenland Ice Sheet enhances melting and subsequent runoff, a major mass loss contributor during the ablation season. The accumulation of both biological (e.g., glacier ice algae) and abiotic (e.g., mineral dust) light-absorbing particulates are important darkening factors, that are potentially influenced by the duration of snow-free, bare ice (a phenological factor), and other geo-topographical factors such as elevation, slope, aspect and the distance from the ice margin. Here, we present the first medium-resolution (30 m) analysis of the phenological and geo-topographical controls on the distribution of dark ice in SE and SW Greenland from statistical analysis of data derived from a harmonized satellite albedo product and ArcticDEM. The duration of bare ice primarily controls the distribution of dark surface ice, allowing for algae growth on inland ice surfaces in particular, whereas geo-topographical factors are only secondary controls.

Plain Language Summary Meltwater that runs off the Greenland Ice Sheet is currently the largest single glaciological contributor to global sea level rise. Recent darkening of the surface ice has contributed significantly to increased melt rates. We used medium resolution (30m) satellite data to link ice surface darkness (albedo) to the topography (elevation, slope, and aspect) of the ice sheet surface, the distance from the ice margin, and the duration of bare or snow-free surface ice. Our analysis shows that the duration of bare ice or snow-free conditions was the most significant factor in driving the darkening of ice. We also found that the impact of topographic factors on the ice darkening varied depending on whether abiotic or biological processes were driving the darkening. This information helps us to better explain where the ice sheet is melting fastest and to predict the loss of ice mass, particularly considering the expected increase in the area of bare, snow-free ice in our warming climate.

1. Introduction

Mass loss from the Greenland Ice Sheet (GrIS) is the largest single cryospheric contributor to sea level rise (Khan et al., 2015; Ryan et al., 2019; Shepherd et al., 2012, 2020). Mass loss has increased sixfold since the 1980s (Mouginot et al., 2019), and accounts for 25% of global mean sea level rise (Chen et al., 2017). Surface albedo is a critical parameter that modulates the absorption of solar radiation by the ice sheet (Box et al., 2012; Naegeli et al., 2015, 2017). Recent darkening of the GrIS enhances surface melt, increases the growth of dark pigmented glacier algae, which further darkens the surface ice and increases melting, referred to as bio-albedo darkening feedback (Chevrollier et al., 2023; Cook et al., 2017, 2020; Stibal et al., 2017). Dark ice is unevenly distributed around the GrIS (Figure 1a), but a band of surface ice with especially low albedo appears annually in the ablation zone of the southwest GrIS between 65° and 70°N, known as the Dark Zone (DZ) (Knap & Oerlemans, 1996; Ryan et al., 2018; Wientjes & Oerlemans, 2010; Wientjes et al., 2017). The positive feedback between meltwater production and darkening accounts for approximately 40% of the annual mass balance change in the DZ (Greuell, 2000). The area of the DZ varies annually (Feng et al., 2017). Bio-albedo feedback is a major darkening mechanism within the DZ (Cook et al., 2013; Tedstone et al., 2017). Bio-albedo feedback is a major darkening mechanism within the DZ (Cook et al., 2020), and is likely to be highly seasonal and stronger in longer summers because of its biological component (Williamson et al., 2018).

Other processes may also explain the darkening of the GrIS and the formation of the DZ. Tedesco et al. (2016) suggested that the darkening of the entire GrIS is driven by increased air temperatures, retreat of the snowline,





Check for updates



Figure 1. Map of the study area (a) and exemplary random sampling points with strict 500 m buffers (b). The map shows the average summer (July–August) snow-free ice albedo ($\alpha_{mean} < 0.65$) in the study period (2009–2017). Contour lines are generated from the ArcticDEM mosaic (Porter et al., 2018). Subfigure c displays the ice sheet area at different elevations in the Southeast and Southwest Greenland Ice Sheet. The dashed lines denote the average snowline (SW:1,550 m a.s.l, SE: 1,453 m a.s.l, adapted from Ryan et al. (2019)).

and accumulation of meltwater (Greuell, 2000; Knap & Oerlemans, 1996) and light-absorbing particulates (LAPs) in the surface ice. These LAPs include dust derived from dry deposition of contemporary atmospheric aerosol, melt out of old dust from outcropping ice (McCutcheon et al., 2021; Wientjes & Oerlemans, 2010; Wientjes et al., 2011, 2012), black carbon, and glacier ice algae (Anesio et al., 2017; Cook et al., 2017, 2020; Lutz et al., 2018; Onuma et al., 2022; Perini et al., 2019; Ryan et al., 2018; Stibal et al., 2017; Williamson et al., 2018; Yallop et al., 2012), and cryoconites (Cook et al., 2018; Takeuchi et al., 2018). The accumulation of LAPs and the bio-albedo feedback amplify ablation and explain more than 70% of the spatial variability of albedo at regional scales (Ryan et al., 2018; Williamson et al., 2018). These studies focus mainly on the DZ, but dark ice is also present in other ablation zones around the GrIS, albeit in more restricted areas (Bøggild et al., 2010). Geo-topographical factors, such as slope and distance from the ice margin, have long been considered potent factors that influence the distribution of dark ice and melt. Terrain also has a strong influence on regulating the incoming solar radiation (J. He et al., 2019), the albedo (Colombo et al., 2023; Picard et al., 2020), and local wind-driven snow transport and melt (Cao & Barros, 2023). Field observations and remote sensing analysis show that sloping, south-facing snow/ice surfaces receive a higher amount of incoming solar radiation (Adhikary et al., 2000; Colombo et al., 2023; J. He et al., 2019) north of the Tropic of Cancer, which also increases surface melt. Finally, the GrIS is fairly flat inland, with steeper slopes near the ice margin, which helps promote katabatic and barrier winds, enhancing surface melt (Wang et al., 2021). The flatter surface ice at higher elevations enhances the accumulation of LAPs and meltwater (Greuell, 2000; Knap & Oerlemans, 1996; Wientjes & Oerlemans, 2010).

The duration of bare, snow-free surface ice is believed to be a key factor influencing the biological darkening through algae growth in summer (Tedstone et al., 2017), and is hereafter referred to as a "phenological" factor. This is strongly correlated with the mean air temperature in July (Shimada et al., 2016) since the average surface albedo is closely associated with the migration of the snowline to higher elevations, which exposes more bare surface ice (Ryan et al., 2019). Here, we aim to investigate whether the regional and local distribution of dark ice in the southeast and the southwest of GrIS can be explained by this simple phenological and other geo-topographical factors (elevation, slope, aspect, and distance from the ice margin) using a harmonized satellite albedo (HSA) product (Feng et al., 2023) and ArcticDEM (Porter et al., 2018).

2. Data and Method

2.1. Data Preprocessing

This study focuses on the southwest (SW) and southeast (SE) regions of the GrIS (Mouginot et al., 2017, 2019; Noël et al., 2019) (Figure 1a). The SW is less cloudy than other Arctic areas because of the persistent high atmospheric pressure (Noël et al., 2019). The increased frequency of anticyclones in central GrIS during the past decades induces more frequent advection of warm southerly winds over the western flank of the GrIS and enhances ablation (Fettweis et al., 2013; Hanna et al., 2014). Surface slopes are also shallower than in the SE (Figure 1). Consequently, the ablation zone in the SW is about 30% larger (Figures 1a and 1c).

ArcticDEM (Porter et al., 2018) provides digital elevation models (DEMs) from 16 August 2009 to 12 March 2017 with 2-m resolution, from which elevation, slope, and aspect were derived. Distance from the ice margin is the geodesic distance from a point on the ice surface to the ice sheet margin, defined by the shapefile of the study area (Mouginot & Rignot, 2019). The HSA (α_{HSA}) is a 30-m resolution albedo product derived from harmonized Landsat and Sentinel 2 data sets (Feng & Cook, 2023; Feng et al., 2023) developed in Google Earth Engine (GEE) (Gorelick et al., 2016). Sentinel 2 surface reflectance data is unavailable on GEE for the study period (2009–2017), so the albedo is mainly derived from Landsat.

Both the ArcticDEMs and the HSA were masked by the GrIS basin data set from Mouginot and Rignot (2019) to remove non-ice areas. Ice surfaces with elevations above 2,000 m a.s.l (green line in Figure 1a) were masked out from our analysis as they are predominantly snow-covered (Fausto & the PROMICE team, 2018). The duration of snow-free, bare ice, hereafter the phenological factor, is derived from the HSA. Typically, ice albedo ranges from 0.20 to 0.65 (e.g., dark ice <0.45, 0.45 < clean ice < 0.55, 0.55 < superimposed ice < 0.65) (Cuffey & Paterson, 2010); and albedo values >0.65 were excluded to limit the analysis to (snow-free) ice surfaces only (Figure 1a). The pixel-wise snow-free duration at the peak of the melt season (July-August), when the annual spatial extent of the DZ is maximum (Tedstone et al., 2017), was calculated from the dates of the first and last day of albedo <0.65. Landsat has a revisiting time of 16 days, and the temporal resolution is 8 days by combining two Landsat satellites (Zhu et al., 2012). Limiting the study to July-August also helps to reduce uncertainties in estimating the phenological factor.

2.2. Random Sampling With Buffer and Filtering

The pre-processed albedo and ArcticDEM were matched by the timestamp (dt < 3 days), and then both data sets were sampled for analysis. The study area was divided into grid cells of 500 m (Figure 1b). Random sampling points (189,857 in the SW and 98,287 in the SE) were generated inside the grid cells with a strict buffer of 500 m to avoid oversampling and spatial autocorrelation, a known issue in geospatial data analysis, which leads to biased results (Gorelick, 2021; Ploton et al., 2020). Sampling points generated at the regional boundary were excluded. Geo-topographical and phenological data were extracted using the generated random sample points at 30-m scale.

2.3. Geo-Topographical Analysis

The filtered and extracted data sets of elevation (m a.s.l), slope (range: $[0, 90]^{\circ}$), aspect (range: $[0, 360]^{\circ}$, where 0° is geographical north), distance from the ice margin (km), and duration of snow-free days were aggregated by year for each sampling point. The ice surface was classified into two classes: (a) dark ice (average albedo in July-August <0.45, number of sampling points: 38,604 in SW, and 9,102 in SE); (b) bare ice (average albedo in July-August ≥ 0.45 and <0.65, number of sampling points: 64,396 in SW, and 27,062 in SE). The definition of the ice classes is somewhat arbitrary, and is determined from spatial patterns of albedo in the DZ (Oerlemans & Vugts, 1993; Wientjes et al., 2011), values used in previous studies (Shimada et al., 2016), the duration of dark ice (Tedstone et al., 2017), and the data density of Landsat imagery (Zhang et al., 2021). We use these empirical values to remain consistent with previous research (Tedstone et al., 2017), but note that albedos <0.45 are primarily driven by the bioalbedo darkening feedback (Cook et al., 2017, 2020; Ryan et al., 2018). A sensitivity

experiment was conducted to assess potential biases arising from superimposed ice (refrozen snowmelt) by restricting bare ice classification to albedos ≥ 0.40 and < 0.55 (Section 4.3).

2.3.1. Regional Comparison and Statistical Analysis

The regional analysis tests the hypothesis that geo-topographical and phenological factors affect the spatial extent of dark ice by analyzing the statistical relationship between albedo and each factor. The Random Forest (RF) classifier (Breiman, 2001) has become increasingly popular in remote sensing because of both its reliability in classification and its capacity to accurately evaluate and rank importance of the predictor (Belgiu & Drăgut, 2016). The averaged annual elevation, slope, aspect, distance, and duration data sets were normalized from 0 to 1 and split into training (70%) and test (30%) data sets for both regions. An RF model was trained to classify the ice types in the SE and SW using the training data set. The relative importance of the factors on ice types in both regions was analyzed by comparing the estimated out-of-bag permuted predictor importance (Breiman et al., 1984; Loh, 2002; Loh & Shih, 1997). The classification accuracy was evaluated using the test data set. A consequence of the random seed selections is that the final results differ even with the same input parameters. Hence, the RF model was run 10 times with the same input, and the resulting predictor importance values were averaged. The model uncertainty range is estimated using one standard deviation around the mean of the 10 runs.

2.3.2. Subregional Analysis in the Southwest GrIS

Dark ice in the SW extends further inland than in the SE (Figure 1a), enabling division of the sampled data into two groups determined by the median distance (Figures 2j and 2l) to the margin, namely margin (distance ≤ 9.57 km) and inland (distance >9.57 km). The number of sampling points for dark and bare ice in the subregional analysis are: (a) bare ice: 23,037 margin and 41,359 inland; (b) dark ice: 19,303 margin and 19,301 inland. Comparison of ice at the margin and inland tests the hypothesis that topographic factors enable the dark ice to extend further inland in the SW than in the SE.

3. Results

3.1. Regional Comparison and Analysis

Dark ice is more prevalent in the SW, with 37.5% of sampled ice pixels being dark ice, compared to only 25.2% in the SE (Figures 2a and 2b). The majority of the SW dark ice (interquartile range, IQR) falls in the elevation range of 673–1,214 m a.s.l (Figure 2c), and the median elevation is 1,013 m a.s.l. This is 500 m higher than the median elevation in the SE (612 m a.s.l.) and >100 m higher than the SE upper quartile (907 m a.s.l). The bare ice median elevation is higher than that of the dark ice in both regions. Generally, slopes in the SE are steeper than in the SW, and the slopes containing dark ice are steeper than those of bare ice (Figure 2f, right-tailed Wilcoxon rank sum test: p < 0.001). The distribution of aspects in the SE is smaller than in the SW (Figure 2i, left-tailed Wilcoxon rank sum test: p < 0.001). The median aspect of dark ice in the SE is 176°, quite close to the geographic south (180°), while the median aspect in the SW is 217°. The distribution of distance from the margin varies greatly in both regions. Half of the dark ice samples in the SE are limited to areas <2.7 km away from the ice margin, and most bare ice surfaces are found no further than 15.3 km (75th percentile) inland (Figures 1a, 2k, and 2l). Dark ice in the SW is also found near the margin (Figure 2j), but extends much further inland (upper whisker >50 km in Figure 2l).

The snow-free duration of bare ice in the SW is statistically longer than in the SE (Figure 2o, right-tailed Wilcoxon rank sum test: p < 0.001), although both the median values for dark ice are 40 days (Figure 2o). The estimated predictor importance of the RF model is shown in Figures 3a–3d. The overall classification accuracies for the SW and SE are 77% and 78%, respectively. The snow-free duration is the most important predictor in both regions, contributing $31 \pm 5\%$ and $33 \pm 3\%$ to the classification of dark and bare ice in the SW and SE, respectively. Elevation is the second most important predictor, with $29 \pm 4\%$ (SW) and $18 \pm 3\%$ (SE), followed by the distance from the ice margin scores ($24 \pm 4\%$ (SW) and $22 \pm 2\%$ (SE)) and slope ($6 \pm 1\%$ (SW) and $15 \pm 2\%$ (SE)).

3.2. Ice at the SW Margin and Inland

The median dark ice elevation is lower than bare ice for both the ice at the margin and inland (Figure 4a). Slopes of dark ice are steeper than bare ice at the margin (Figure 4b, right-tailed Wilcoxon rank sum test: p < 0.001).



Geophysical Research Letters

10.1029/2023GL104894



Figure 2. Associations between (a and b) albedo and elevation, (d and e) slope, (g and h) aspect, (j and k) distance from ice margin, and (m and n) snow-free duration. The blue dashed lines indicate the threshold (albedo = 0.45) separating bare and dark ice. The distributions of each factor are shown in the notched boxplots in the column to the right (c, f, i, l, o), respectively. Ice classes are indicated by colors in the boxplots.

However, the inland ice surfaces are mostly flat (IQR between 1 and 2°), regardless of the ice class. The distribution of aspect at the margin and inland differ (Figure 4c). Dark ice aspects are statistically smaller than for bare ice at the margin but are greater inland (one-sided Wilcoxon rank sum test: p < 0.001). The median of the aspect of inland dark ice is 235°, which is closer to the geographic west than the dark ice near the margin (median = 193°). The snow-free duration of dark ice near the margin is typically longer than the inland ice (Figure 4d).

4. Discussion

4.1. Geo-Topographical and Phenological Controls on Dark Ice in the SW and SE

The snow-free duration is the most critical factor (Figure 3) in the occurrence of dark ice in both regions (median = 40 days, Figures 2m-2o). Longer durations give the biological component of LAPs, glacial ice algae,



Geophysical Research Letters



Figure 3. Out-of-bag permuted predictor importance estimated from the Random Forest model for geo-topographical and phenological factors in the (a and b) SW and (c and d) SE. The errorbars show the standard deviation of the predictor importance.

more time to grow. Meltwater and the abiotic components of LAPs (dust, cryoconite, and black carbon) may also accumulate as the melt season progresses (Lloyd, 2021; Williamson et al., 2019, 2020). We hypothesize that the presence of algae helps to aggregate and retain abiotic impurities due to their production of cohesive compounds, such as extracellular polymeric substances (Holland et al., 2019; Stibal et al., 2012), which in turn reduce surface albedo and generate more melting. Snowlines are projected to retreat into the flatter ice sheet interior (Figures 1a and 1c) under warmer climates, unless offset by enhanced winter accumulation (Ryan et al., 2019). Our analysis also suggests that future expansion of the bare ice area and prolonged snow-free duration may induce a rapid expansion of dark ice distribution, causing positive feedback resulting in even more surface melt.

The spatial distribution of dark ice is limited to the ice margin in the SE, while the dark ice can extend up to 50 km inland in the SW (Figures 1a and 2j–2l). The areas below the average snowline (Ryan et al., 2019) make up about 23.6% and 21.8% of the total area of the SW and SE, respectively (Figure 1c). However, the dark ice to bare ice ratio in these regions is disproportionally different, with 37.5% in the SW and 25.2% in the SE. The snow-free duration of bare ice in the SW is statistically longer than in the SE (Figure 2o), making it possible for microbial communities to colonize more ice surface.

Geo-topographical factors play a secondary role in regulating the distribution of dark ice, although elevation, surface slope and aspect may influence the snow-free duration. Generally, the snow-free duration decreases with elevation, but we note that dark ice in the SW occurs at higher elevations than in the SE. South-facing (median = 179°) slopes predominate in the dark ice of the SE. The influence of LAPs is more crucial on south-facing slopes than on flatter surfaces because the ice receives more intense solar radiation (Adhikary et al., 2000; Colombo et al., 2023); therefore, the snow-free duration is prolonged.

4.2. Geo-Topographical and Phenological Controls on Dark Ice at the Margin and Inland in SW

The bare ice and dark ice of the ice margin and inland in the SW have different relations with topographic factors (Figure 4). The slopes of dark ice are statistically steeper at the margin than inland (Figures 2d–2f and 4b).



Figure 4. Distribution and summary statistics (boxenplots Waskom (2021)) for (a) elevation, (b) slope, (c) aspect, and (d) snow-free duration in the southwest margin (distance \leq 9.57 km) and inland (distance >9.57 km) region. Ice classes are indicated by color.

Steeper slopes near the ice margin expose bare ice at the onset of spring if winter accumulation is small (van den Broeke et al., 2009); therefore, the snow-free duration is longer (median duration = 47 days, Figure 4d). Melting downslope can be further enhanced by katabatic winds (Wang et al., 2021). However, the influence of slope is small across inland GrIS, since surface slopes are generally low (over 75% of the interior area $<2^{\circ}$).

Dark ice is more dependent on the snow-free duration inland than at the margin, where snow-free periods are longer (Figure 4d). Aspect is a key parameter that adjusts the solar irradiance (Gutiérrez-Jurado & Vivoni, 2013; Rogeau, 2018), and dark ice aspects at the margin are closer to geographic south than bare ice (Figure 4c), as is also found in the SE (Figures 2h and 2i). However, inland dark ice favors aspects (median = 235°) that deviate further from the geographic south. This is because the darkening of the DZ is strongly associated with the growth of glacier ice algae (Cook et al., 2017, 2020), which can be suppressed under high light intensity (Jensen et al., 2023). The quantum efficiency of glacier algal photosystem II (PSII) is depressed under higher irradiances found on the GrIS (Williamson et al., 2020), and it could be that southwesterly aspects favor glacier algal growth over direct south-facing areas that receive the greatest irradiance. The more westerly facing flat ice surfaces in the SW may promote glacier algae colonization of areas with less intensive radiation.

4.3. Limitations

The definition of dark ice relies on an arbitrary albedo threshold ($\alpha < 0.45$) and the duration that it remains dark. Future work should quantify the contributions of biological and abiotic darkening factors to dark ice formation. The snow-free duration was determined from the HSA. The accuracy is subject to the temporal resolution of clear (cloud- and saturation-free) observations (Zhang et al., 2021). The revisiting time of two Landsats is 8 days at best; therefore, it may not capture the exact timing of the onset and end of the melt season. The impact of this uncertainty is assumed to be limited as we restricted the analysis to the peak of the melt season (July-August). Saturated snow has an albedo <0.65 (Cuffey & Paterson, 2010), which may increase noise in the analysis. However, this higher threshold helps to capture the duration of melt season using low temporal resolution satellite data. A sensitivity experiment was run by changing the dark ice upper albedo threshold from 0.65 to 0.55 and the lower from 0.45 to 0.40. These changes do not have a great impact on the main statistical results (Table S1 in Supporting Information S1) or the rankings of the relative predictor importance (Figures S1 and S2 in Supporting Information S1). The difference between the estimated predictor importance of elevation (16.2 ± 2.4) and duration (16.3 ± 2.9) after changing the bare ice threshold to 0.55 (Figure S1 in Supporting Information S1) is very minor compared to Figure 3. Hence, the choice of dark ice thresholds in the ranges we have chosen does not change our main conclusions.

5. Conclusion

The darkening of ice on the GrIS results from a complex combination of abiotic and biological processes. Geo-topographical factors (elevation, slope, aspect, and the distance from the ice margin) and the duration of snow-free bare ice, a phenological factor were analyzed at regional and subregional scales. Results revealed that the snow-free duration is the primary factor governing the darkening of surface ice, enabling LAPs and meltwater to accumulate. Climate forcings regulate the extent of snow-free ablation areas, making room for new bare ice to darken. Geo-topographical factors are secondary controls on the darkening processes at an annual resolution. Surface slope and aspect influence the darkening by modulating the snow-free duration in the SE and the margin of the SW, where the dark ice predominates on steeper slopes and south-facing aspects. We hypothesize that aspect influences the biological darkening of the inland flat surface ice, where southwesterly slopes may be preferentially colonized over southerly slopes because of their lower solar irradiation receipts. Future upward migration of snowlines and longer snow-free duration may trigger a rapid expansion of dark ice on the GrIS, causing positive feedback that amplifies meltwater runoff production.

Data Availability Statement

The scripts for data processing and analysis are available at Feng (2023a). Data are available on GEE (Gorelick et al., 2016) and processed mainly via geemap (Wu, 2020; Wu et al., 2019). A story map of the topographic factors in the study area is accessible at Feng (2023b).



Acknowledgments

This publication is part of the Deep Purple Project, funded by the European Research Council under the European Union's Horizon 2020 research and innovation program (Grant agreement No 856416). Kathrin Naegeli is supported by the ESA PRODEX Trishna T-SEC project (PEA C4000133711). DEMs were provided by the Polar Geospatial Center under NSF-OPP awards 1043681, 1559691, and 1542736. The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper. We would like to express our thanks to an anonymous reviewer, Chris Williamson. and the editor, Mathieu Morlighem, for their insightful commentary, which greatly improved the quality of this manuscript. Ian Stevens made valuable suggestions during the revision process.

References

- Adhikary, S., Nakawo, M., & Seko, K. (2000). Dust influence on the melting process of glacier ice: Experimental results from Lirung Glacier, Nepal Himalayas. In IAHS-AISH Publication (pp. 43–52). IAHS Publication.
- Anesio, A. M., Lutz, S., Chrismas, N. A., & Benning, L. G. (2017). The microbiome of glaciers and ice sheets. *npj Biofilms and Microbiomes*, 3(1), 10. https://doi.org/10.1038/s41522-017-0019-0
- Belgiu, M., & Drăguţ, L. (2016). Random forest in remote sensing: A review of applications and future directions. ISPRS Journal of Photogrammetry and Remote Sensing, 114, 24–31. https://doi.org/10.1016/j.isprsjprs.2016.01.011
- Bøggild, C. E., Brandt, R. E., Brown, K. J., & Warren, S. G. (2010). The ablation zone in northeast Greenland: Ice types, albedos and impurities. Journal of Glaciology, 56(195), 101–113. https://doi.org/10.3189/002214310791190776
- Box, J. E., Fettweis, X., Stroeve, J. C., Tedesco, M., Hall, D. K., & Steffen, K. (2012). Greenland ice sheet albedo feedback: Thermodynamics and atmospheric drivers. *The Cryosphere*, 6(4), 821–839. https://doi.org/10.5194/tc-6-821-2012
- Breiman, L. (2001). Random Forests. Machine Learning, 45(1), 5-32. https://doi.org/10.1023/A:1010933404324
- Breiman, L., Friedman, J., Olshen, R., & Stone, C. (1984). Classification and regression trees. CRC Press.
 - Cao, Y., & Barros, A. P. (2023). Topographic controls on active microwave behavior of mountain snowpacks. *Remote Sensing of Environment*, 284, 113373. https://doi.org/10.1016/j.rse.2022.113373
- Chen, X., Zhang, X., Church, J. A., Watson, C. S., King, M. A., Monselesan, D., et al. (2017). The increasing rate of global mean sea-level rise during 1993–2014. *Nature Climate Change*, 7(7), 492–495. https://doi.org/10.1038/nclimate3325
- Chevrollier, L.-A., Cook, J. M., Halbach, L., Jakobsen, H., Benning, L. G., Anesio, A. M., & Tranter, M. (2023). Light absorption and albedo reduction by pigmented microalgae on snow and ice. *Journal of Glaciology*, 69(274), 333–341. https://doi.org/10.1017/jog.2022.64
- Colombo, R., Pennati, G., Pozzi, G., Garzonio, R., Di Mauro, B., Giardino, C., et al. (2023). Mapping snow density through thermal inertia observations. *Remote Sensing of Environment*, 284, 113323. https://doi.org/10.1016/j.rse.2022.113323
- Cook, J. M., Hodson, A. J., Gardner, A. S., Flanner, M., Tedstone, A. J., Williamson, C., et al. (2017). Quantifying bioalbedo: A new physically based model and discussion of empirical methods for characterising biological influence on ice and snow albedo. *The Cryosphere*, 11(6), 2611–2632. https://doi.org/10.5194/tc-11-2611-2017
- Cook, J. M., Sweet, M., Cavalli, O., Taggart, A., & Edwards, A. (2018). Topographic shading influences cryoconite morphodynamics and carbon exchange. Arctic, Antarctic, and Alpine Research, 50(1), S100014. https://doi.org/10.1080/15230430.2017.1414463
- Cook, J. M., Tedstone, A. J., Williamson, C., McCutcheon, J., Hodson, A. J., Dayal, A., et al. (2020). Glacier algae accelerate melt rates on the south-western Greenland Ice Sheet. *The Cryosphere*, 14(1), 309–330. https://doi.org/10.5194/tc-14-309-2020
- Cuffey, K. M., & Paterson, W. S. B. (2010). The physics of glaciers. Elsevier Science & Technology Books.
- Fausto, R. S., & the PROMICE team. (2018). The Greenland ice sheet Snowline elevations at the end of the melt seasons from 2000 to 2017. Geological Survey of Denmark and Greenland Bulletin, 41, 71–74. https://doi.org/10.34194/geusb.v41.4346
- Feng, S. (2023a). fsn1995/topography-control-of-ice-albedo: Geo-topographic and phenological analysis of the darkening of the Greenland Ice Sheet (v1.0) [Dataset]. Zenodo. https://doi.org/10.5281/ZENODO.8016712
- Feng, S. (2023b). Geo-topographical and phenological controls on ice darkening. Retrieved from https://storymaps.arcgis.com/stories/ f52a6a97b8eb4c76af56d848dd5330bb
- Feng, S., & Cook, J. M. (2023). Remote-Sensing-of-Albedo: Development of harmonized satellite albedo [Dataset]. Zenodo. https://doi. org/10.5281/zenodo.7642574
- Feng, S., Cook, J. M., Anesio, A. M., Benning, L. G., & Tranter, M. (2023). Long time series (1984–2020) of albedo variations on the Greenland ice sheet from harmonized Landsat and Sentinel 2 imagery. *Journal of Glaciology*, 69(277), 1–16. https://doi.org/10.1017/jog.2023.11
- Fettweis, X., Hanna, E., Lang, C., Belleflamme, A., Erpicum, M., & Gallée, H. (2013). Important role of the mid-tropospheric atmospheric circulation in the recent surface melt increase over the Greenland ice sheet. *The Cryosphere*, 7(1), 241–248. https://doi.org/10.5194/tc-7-241-2013
- Gorelick, N. (2021). Random samples with buffering. Earth Engine by Example. Retrieved from https://medium.com/google-earth/ random-samples-with-buffering-6c8737384f8c
- Gorelick, N., Hancher, M., Dixon, M., Ilyushchenko, S., Thau, D., & Moore, R. (2016). Google Earth Engine: Planetary-scale geospatial analysis for everyone. *Remote Sensing of Environment*, 202, 18–27. https://doi.org/10.1016/j.rse.2017.06.031
- Greuell, W. (2000). Melt–water accumulation on the surface of the Greenland ice sheet: Effect on albedo and mass balance. *Geografiska Annaler Series A: Physical Geography*, 82(4), 489–498. https://doi.org/10.1111/j.0435-3676.2000.00136.x
- Gutiérrez-Jurado, H. A., & Vivoni, E. R. (2013). Ecogeomorphic expressions of an aspect-controlled semiarid basin: II. Topographic and vegetation controls on solar irradiance. *Ecohydrology*, 6(1), 24–37. https://doi.org/10.1002/eco.1263
- Hanna, E., Fettweis, X., Mernild, S. H., Cappelen, J., Ribergaard, M. H., Shuman, C. A., et al. (2014). Atmospheric and oceanic climate forcing of the exceptional Greenland ice sheet surface melt in summer 2012. *International Journal of Climatology*, 34(4), 1022–1037. https://doi. org/10.1002/joc.3743
- He, J., Zhao, W., Li, A., Wen, F., & Yu, D. (2019). The impact of the terrain effect on land surface temperature variation based on Landsat-8 observations in mountainous areas. *International Journal of Remote Sensing*, 40(5–6), 1808–1827. https://doi.org/10.1080/01431161.2018.1466082
- He, T., Liang, S., Yu, Y., Wang, D., Gao, F., & Liu, Q. (2013). Greenland surface albedo changes in July 1981–2012 from satellite observations. *Environmental Research Letters*, 8(4), 044043. https://doi.org/10.1088/1748-9326/8/4/044043
- Holland, A. T., Williamson, C. J., Sgouridis, F., Tedstone, A. J., McCutcheon, J., Cook, J. M., et al. (2019). Dissolved organic nutrients dominate melting surface ice of the Dark Zone (Greenland Ice Sheet). *Biogeosciences*, 16(16), 3283–3296. https://doi.org/10.5194/bg-16-3283-2019
- Jensen, M. B., Perini, L., Halbach, L., Jakobsen, H., Haraguchi, L., Ribeiro, S., et al. (2023). The dark art of cultivating glacier ice algae. *Botany Letters*, *1*, 1–10. https://doi.org/10.1080/23818107.2023.2248235
- Khan, S. A., Aschwanden, A., Bjørk, A. A., Wahr, J., Kjeldsen, K. K., & Kjær, K. H. (2015). Greenland ice sheet mass balance: A review. Reports on Progress in Physics, 78(4), 046801. https://doi.org/10.1088/0034-4885/78/4/046801
- Knap, W. H., & Oerlemans, J. (1996). The surface albedo of the Greenland ice sheet: Satellite-derived and in situ measurements in the Søndre Strømfjord area during the 1991 melt season. *Journal of Glaciology*, 42(141), 364–374. https://doi.org/10.3189/S0022143000004214

Lloyd, K. G. (2021). Time as a microbial resource. *Environmental Microbiology Reports*, *13*(1), 18–21. https://doi.org/10.1111/1758-2229.12892 Loh, W.-Y. (2002). Regression tress with unbiased variable selection and interaction detection. *Statistica Sinica*, 361–386.

- Loh, W.-Y., & Shih, Y.-S. (1997). Split selection methods for classification trees. Statistica Sinica, 815-840.
- Lutz, S., McCutcheon, J., McQuaid, J. B., & Benning, L. G. (2018). The diversity of ice algal communities on the Greenland Ice Sheet as revealed by oligotyping. *Microbial Genomics*, 4(3), e000159. https://doi.org/10.1099/mgen.0.000159
- McCutcheon, J., Lutz, S., Williamson, C., Cook, J. M., Tedstone, A. J., Vanderstraeten, A., et al. (2021). Mineral phosphorus drives glacier algal blooms on the Greenland Ice Sheet. *Nature Communications*, 12(1), 570. https://doi.org/10.1038/s41467-020-20627-w

Mouginot, J., & Rignot, E. (2019). Glacier catchments/basins for the Greenland Ice Sheet [Dataset]. Dryad. https://doi.org/10.7280/D1WT11 Mouginot, J., Rignot, E., Bjørk, A. A., van den Broeke, M., Millan, R., Morlighem, M., et al. (2019). Forty-six years of Greenland Ice Sheet mass balance from 1972 to 2018. Proceedings of the National Academy of Sciences, 116(19), 9239–9244. https://doi.org/10.1073/

pnas.1904242116 Mouginot, J., Rignot, E., Scheuchl, B., & Millan, R. (2017). Comprehensive annual ice sheet velocity mapping using Landsat-8, Sentinel-1, and RADARSAT-2 data. *Remote Sensing*, 9(4), 364. https://doi.org/10.3390/rs9040364

- Naegeli, K., Damm, A., Huss, M., Schaepman, M., & Hoelzle, M. (2015). Imaging spectroscopy to assess the composition of ice surface materials and their impact on glacier mass balance. *Remote Sensing of Environment*, 168, 388–402. https://doi.org/10.1016/j.rse.2015.07.006
- Naegeli, K., Damm, A., Huss, M., Wulf, H., Schaepman, M., & Hoelzle, M. (2017). Cross-Comparison of albedo products for glacier surfaces derived from airborne and satellite (Sentinel-2 and Landsat 8) optical data. *Remote Sensing*, 9(2), 110. https://doi.org/10.3390/ rs9020110
- Noël, B., van de Berg, W. J., Lhermitte, S., & van den Broeke, M. R. (2019). Rapid ablation zone expansion amplifies north Greenland mass loss. Science Advances, 5(9), eaaw0123. https://doi.org/10.1126/sciadv.aaw0123
- Oerlemans, J., & Vugts, H. F. (1993). A meteorological experiment in the melting zone of the Greenland Ice Sheet. Bulletin of the American Meteorological Society, 74(3), 355–366. https://doi.org/10.1175/1520-0477(1993)074
- Onuma, Y., Takeuchi, N., Uetake, J., Niwano, M., Tanaka, S., Nagatsuka, N., & Aoki, T. (2022). Modeling seasonal growth of phototrophs on bare ice on the Qaanaaq Ice Cap, northwestern Greenland. *Journal of Glaciology*, 69(275), 1–13. https://doi.org/10.1017/jog.2022.76
- Perini, L., Gostinčar, C., Anesio, A. M., Williamson, C., Tranter, M., & Gunde-Cimerman, N. (2019). Darkening of the Greenland Ice Sheet: Fungal abundance and diversity are associated with algal bloom. *Frontiers in Microbiology*, 10, 557. https://doi.org/10.3389/fmicb.2019. 00557
- Picard, G., Dumont, M., Lamare, M., Tuzet, F., Larue, F., Pirazzini, R., & Arnaud, L. (2020). Spectral albedo measurements over snow-covered slopes: Theory and slope effect corrections. *The Cryosphere*, 14(5), 1497–1517. https://doi.org/10.5194/tc-14-1497-2020
- Ploton, P., Mortier, F., Réjou-Méchain, M., Barbier, N., Picard, N., Rossi, V., et al. (2020). Spatial validation reveals poor predictive performance of large-scale ecological mapping models. *Nature Communications*, 11(1), 4540. https://doi.org/10.1038/s41467-020-18321-y
- Porter, C., Morin, P., Howat, I., Noh, M.-J., Bates, B., Peterman, K., et al. (2018). ArcticDEM. https://doi.org/10.7910/DVN/OHHUKH Rogeau, M.-P. (2018). Aspect. In *Encyclopedia of wildfires and wildland-urban interface (WUI) fires* (pp. 1–3). Springer International Publish-
- ing. https://doi.org/10.1007/978-3-319-51727-8{_}109-1
 Ryan, J. C., Hubbard, A., Stibal, M., Irvine-Fynn, T. D., Cook, J., Smith, L. C., et al. (2018). Dark zone of the Greenland Ice Sheet controlled by distributed biologically-active impurities. *Nature Communications*, 9(1), 1065. https://doi.org/10.1038/s41467-018-03353-2
- Ryan, J. C., Smith, L. C., van As, D., Cooley, S. W., Cooper, M. G., Pitcher, L. H., & Hubbard, A. (2019). Greenland Ice Sheet surface melt
- amplified by snowline migration and bare ice exposure. *Science Advances*, 5(3), eaav3738. https://doi.org/10.1126/sciadv.aav3738 Shepherd, A., Ivins, E., Rignot, E., Smith, B., van den Broeke, M., Velicogna, I., et al. (2020). Mass balance of the Greenland Ice Sheet from 1992
- to 2018. *Nature*, 579(7798), 233–239. https://doi.org/10.1038/s41586-019-1855-2 Shepherd, A., Ivins, E. R., Geruo, A., Barletta, V. R., Bentley, M. J., Bettadpur, S., et al. (2012). A reconciled estimate of ice-sheet mass balance. *Science*, 338(6111), 1183–1189. https://doi.org/10.1126/science.1228102
- Shimada, R., Takeuchi, N., & Aoki, T. (2016). Inter-annual and geographical variations in the extent of bare ice and dark ice on the Greenland Ice Sheet derived from MODIS satellite images. Frontiers in Earth Science, 4, 43. https://doi.org/10.3389/feart.2016.00043
- Stibal, M., Box, J. E., Cameron, K. A., Langen, P. L., Yallop, M. L., Mottram, R. H., et al. (2017). Algae drive enhanced darkening of bare ice on the Greenland Ice Sheet. *Geophysical Research Letters*, 44(22), 11463–11471. https://doi.org/10.1002/2017GL075958
- Stibal, M., Šabacká, M., & Žárský, J. (2012). Biological processes on glacier and ice sheet surfaces. Nature Geoscience, 5(11), 771–774. https:// doi.org/10.1038/ngeo1611
- Takeuchi, N., Sakaki, R., Uetake, J., Nagatsuka, N., Shimada, R., Niwano, M., & Aoki, T. (2018). Temporal variations of cryoconite holes and cryoconite coverage on the ablation ice surface of Qaanaaq Glacier in northwest Greenland. *Annals of Glaciology*, 59(77), 21–30. https://doi. org/10.1017/aog.2018.19
- Tedesco, M., Doherty, S., Fettweis, X., Alexander, P., Jeyaratnam, J., & Stroeve, J. C. (2016). The darkening of the Greenland ice sheet: Trends, drivers, and projections (1981–2100). The Cryosphere, 10(2), 477–496. https://doi.org/10.5194/tc-10-477-2016
- Tedstone, A. J., Bamber, J. L., Cook, J. M., Williamson, C. J., Fettweis, X., Hodson, A. J., & Tranter, M. (2017). Dark ice dynamics of the southwest Greenland Ice Sheet. *The Cryosphere*, 11(6), 2491–2506. https://doi.org/10.5194/tc-11-2491-2017
- van den Broeke, M., Smeets, P., & Ettema, J. (2009). Surface layer climate and turbulent exchange in the ablation zone of the west Greenland ice sheet. *International Journal of Climatology*, 29(15), 2309–2323. https://doi.org/10.1002/joc.1815
- Wang, W., Zender, C. S., van As, D., Fausto, R. S., & Laffin, M. K. (2021). Greenland surface melt dominated by solar and sensible heating. *Geophysical Research Letters*, 48(7), e2020GL090653. https://doi.org/10.1029/2020GL090653
- Waskom, M. (2021). seaborn: Statistical data visualization. Journal of Open Source Software, 6(60), 3021. https://doi.org/10.21105/joss. 03021
- Wientjes, I., & Oerlemans, J. (2010). An explanation for the dark region in the western melt zone of the Greenland ice sheet. *The Cryosphere*, 4(3), 261–268. https://doi.org/10.5194/tc-4-261-2010
- Wientjes, I., Van de Wal, R., Reichart, G. J., Sluijs, A., & Oerlemans, J. (2011). Dust from the dark region in the western ablation zone of the Greenland ice sheet. *The Cryosphere*, 5(3), 589–601. https://doi.org/10.5194/tc-5-589-2011
- Wientjes, I., Van De Wal, R., Schwikowski, M., Zapf, A., Fahrni, S., & Wacker, L. (2012). Carbonaceous particles reveal that Late Holocene dust causes the dark region in the western ablation zone of the Greenland ice sheet. *Journal of Glaciology*, 58(210), 787–794. https://doi. org/10.3189/2012JoG11J165
- Williamson, C. J., Anesio, A. M., Cook, J., Tedstone, A., Poniecka, E., Holland, A., et al. (2018). Ice algal bloom development on the surface of the Greenland Ice Sheet. *FEMS Microbiology Ecology*, 94(3), 25. https://doi.org/10.1093/femsec/fiy025
- Williamson, C. J., Cameron, K. A., Cook, J. M., Zarsky, J. D., Stibal, M., & Edwards, A. (2019). Glacier algae: A dark past and a darker future. Frontiers in Microbiology, 10, 519. https://doi.org/10.3389/fmicb.2019.00524
- Williamson, C. J., Cook, J., Tedstone, A., Yallop, M., McCutcheon, J., Poniecka, E., et al. (2020). Algal photophysiology drives darkening and melt of the Greenland Ice Sheet. Proceedings of the National Academy of Sciences, 117(11), 5694–5705. https://doi.org/10.1073/pnas. 1918412117
- Wu, Q. (2020). geemap: A Python package for interactive mapping with Google Earth Engine. Journal of Open Source Software, 5(51), 2305. https://doi.org/10.21105/joss.02305
- Wu, Q., Lane, C. R., Li, X., Zhao, K., Zhou, Y., Clinton, N., et al. (2019). Integrating LiDAR data and multi-temporal aerial imagery to map wetland inundation dynamics using Google Earth Engine. *Remote Sensing of Environment*, 228, 1–13. https://doi.org/10.1016/j.rse.2019.04.015

- Yallop, M. L., Anesio, A. M., Perkins, R. G., Cook, J., Telling, J., Fagan, D., et al. (2012). Photophysiology and albedo-changing potential of the ice algal community on the surface of the Greenland ice sheet. *The ISME Journal*, 6(12), 2302–2313. https://doi.org/10.1038/ismej.2012.107
 Zhang, J., Shang, R., Rittenhouse, C., Witharana, C., & Zhu, Z. (2021). Evaluating the impacts of models, data density and irregularity on reconstructing and forecasting dense Landsat time series. *Science of Remote Sensing*, *4*, 100023. https://doi.org/10.1016/j.srs.2021.100023
- Zhu, Z., Woodcock, C. E., & Olofsson, P. (2012). Continuous monitoring of forest disturbance using all available Landsat imagery. Remote Sensing of Environment, 122, 75–91. https://doi.org/10.1016/j.rse.2011.10.030